4-18

Boyce-Codd Normal Form

Third Normal Form is defined with reference to a selected

primary key. Functional dependencies may be broken even

though a schema is maintained in 3NF. For example:

Sample_Schema

A B C D X Y

77 smith alf pat megan 371

65 smith jim eva fred 83

47 smith jim ada bob 197

Here we assume that the primary key attributes are {A, B}

and that the attribute set {B, C, D} is also a candidate key.

It is possible for there to be a FD {B, C} fi X without

breaking 3NF, since this dependency is on the values of a

pair of {key, non-key} attributes. (NO, I've not come up

with a convincing story to account for these dependencies)

How can we fix this gap, and hopefully establish FD-based

criteria which do not depend on arbitrary choice of key?